Author's response to reviews

Title: Preoperative neutrophil-to-lymphocyte ratio is an independent prognostic marker in patients with laryngeal squamous cell carcinoma

Authors:

Xiuping Tu (tuxiuping105@163.com)
Qianhui Qiu (qiugqianhui@hotmail.com)
Liangsi Chen (lancer_chen@hotmail.com)
Xiaoning Luo (luoxn_ent@126.com)
Zhongming Lu (entlu@126.com)
Siyi Zhang (syzhang555@hotmail.com)
Shaohua Chen (chenshaohua1168@163.com)

Version: 6 Date: 5 September 2015

Author's response to reviews:

Dear Editors:

We thank you consider publication concerning our manuscript entitled “Preoperative neutrophil-to-lymphocyte ratio is an independent prognostic marker in patients with laryngeal squamous cell carcinoma” (MS: 1755169606157600).

We have made correction according to your suggestions. The responses to those comments are as listed as follows.

Editor's Additional Request:
1. In Table 2 survival is in two measures, broken down by gender. There are only 4 females. This level of detail may allow the participants to be identified. Please alter table 2 to remove the breakdown by gender.

Response # Thank you for your advice. We have removed the breakdown by gender in the Table2. (Line 512, Page24)

2. Introduction to be renamed Background.

Response # We have revised it in the paper. Thank you. (Line 67, Page4)